Author's response to reviews

Title: Intraoperative and fluorescein angiographic findings of a secondary macular hole associated with age-related macular degeneration treated by pars plana vitrectomy

Authors:

Tomohiro Okamoto (pettenc@hotmail.com)
Hajime Shinoda (shinoha@mac.com)
Toshihide Kurihara (kuriboh2@yahoo.co.jp)
Norihiro Nagai (nagai@a5.keio.jp)
Tsubota Tsubota (tsubota@z3.keio.jp)
Ozawa Ozawa (ozawa@a5.keio.jp)

Version: 2  Date: 12 July 2014

Author's response to reviews:

Dear Editor,

According to the guide of the editorial office, we added the line numbering and e-mail address of all the co-authors in the text.

Thank you for your cooperation.

Sincerely yours,

Yoko Ozawa M.D., Ph.D.
Assistant Professor
Lab Chief
Laboratory of Retinal Cell Biology
Department of Ophthalmology
Keio University, School of Medicine
35 Shinanomachi
Shinjukuku Tokyo 160-8582 Japan
TEL; +81333531211 ext.62402
+81353633869
FAX; +81353633869

e-mail; ozawa@a5.keio.jp